

UvA-DARE (Digital Academic Repository)

A panel data analysis of the effects of wages, standard hours and unionisation on paid overtime work in Britain"

Kalwij, A.S.; Gregory, M.

Publication date 2004

Link to publication

Citation for published version (APA):

Kalwij, A. S., & Gregory, M. (2004). *A panel data analysis of the effects of wages, standard hours and unionisation on paid overtime work in Britain*". (AIAS Working Papers; No. 2003/22). Unknown Publisher.

General rights

It is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), other than for strictly personal, individual use, unless the work is under an open content license (like Creative Commons).

Disclaimer/Complaints regulations

If you believe that digital publication of certain material infringes any of your rights or (privacy) interests, please let the Library know, stating your reasons. In case of a legitimate complaint, the Library will make the material inaccessible and/or remove it from the website. Please Ask the Library: https://uba.uva.nl/en/contact, or a letter to: Library of the University of Amsterdam, Secretariat, Singel 425, 1012 WP Amsterdam, The Netherlands. You will be contacted as soon as possible.

Universiteit van Amsterdam



Amsterdam Institute for Advanced labour Studies

A PANEL DATA ANALYSIS OF THE EFFECTS OF WAGES,

STANDARD HOURS AND UNIONISATION

ON PAID OVERTIME WORK IN BRITAIN

Adriaan S. Kalwij

Tilburg University

Mary Gregory

University of Oxford

Working Paper 2003-22 February 2004

Acknowledgements

Financial support from the Leverhulme Trust project "The labour market consequences of technical and structural change" is gratefully acknowledged. We wish to thank the seminar participants at the Leverhulme Trust project meeting at the University of Essex, the 2000 Annual Conference of the Royal Economic Society at the University of St. Andrews, the Centre for Economic Performance (CEP) at the London School of Economics, the Institute for the Study of Labor (IZA) in Bonn, the EALE-SOLE joint meeting in Milan, the Applied Econometrics Association in Helsinki, the Tinbergen Institute at the University of Amsterdam, University of Cardiff and the Labour Economics seminar at the University of Oxford. The editor Peter Lynn, the associate editor and two anonymous referees provided extremely useful comments and discussion.

Information for library

Adriaan S. Kalwij (2003), Working Papers Series, Amsterdam Institute for Advanced Labour Studies, paper 2003-22. Amsterdam: University of Amsterdam

Contact: Room B933, Department of Economics, Tilburg University,

PO Box 90153, 5000 LE Tilburg, The Netherlands. (Email: kalwij@ uvt.nl)

Keywords: Panel data, Censoring, Overtime work

JEL-codes: C23, C24, J2October 2003

© All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted, in any form, or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior permission of the auteur.

© Adriaan S. Kalwij & Mary Gregory, Amsterdam, October, 2003 This paper can be downloaded: www.uva-aias.net/files/working_papers/WP22.pdf A PANEL DATA ANALYSIS OF THE EFFECTS OF WAGES, STANDARD HOURS AND UNIONISATION

ON PAID OVERTIME WORK IN BRITAIN

ABSTRACT

This study examines the effects of the basic wage rate, standard working hours and unionisation on paid overtime work in Britain using individual-level data from the New Earnings Survey over the period 1975-2001. For this purpose we estimate a panel data model. We show that to obtain consistent estimates it is important to allow for both the censoring of paid overtime hours at zero and for correlations between the explanatory variables and unobserved individual specific effects. The main empirical results are that a reduction in standard hours increases both overtime incidence and overtime hours, while an increase in the wage rate decreases the incidence of overtime but brings a small increase in overtime hours for those working overtime. For men the effects are stronger than for women. Union coverage is of minor empirical importance. The occupation and industry structure of employment has shifted from high to lower overtime jobs. Taken together, these economic variables can explain almost half of the changing incidence of overtime for men, and most of the change in overtime hours worked by women, but are less successful in explaining the changes in overtime hours worked by men or the incidence of overtime for women.

TABLE OF CONTENTS

I		I
2	THE NEW EARNINGS SURVEY 1975-2001	7
2.1	Definitions and data selection	7
2.2	Panel attrition	9
2.3	Descriptive statistics	_ 11
3		_ 13
3.1	Overtime incidence	_ 13
3.2	Overtime hours	_ 14
4		_ 19
4.1	Overtime incidence	_ 19
4.2	Overtime hours	_ 23
4.3	Predictions	_ 28
5		_ 32
Ref		_ 34
Арр		_ 36
Арр	PENDIX B: OVERTIME INCIDENCE, OVERTIME HOURS (IF WORKING OVERTIME), AND THE DISTRIBUTION OF EMPLOYMENT BY OCCUPATION, INDUSTRY AND REGION	_ 39

I INTRODUCTION

An important determinant of the level of employment is the division of any increase in demand for labour services between an increase in hours worked and an expansion of employment. While the evolution of employment has received intensive scrutiny in both analysis and policy-making, much less attention has been given to the intensive margin of labour adjustment, i.e. the use of overtime hours. Currently in the Britain paid overtime is worked by one-third of male employees, 7-9 hours per week on average, and by almost one-fifth of female employees, 3-4 hours per week on average (see Table I below). These figures are in line with other observations for Britain by Bell and Hart (1998), and for the US (Trejo, 1993) and Germany (Bauer and Zimmermann, 1999). This shows that paid overtime makes up a significant proportion of hours worked, and consequently of the labour costs of firms and the labour income of households.

The economic analysis of overtime hours has been extensively developed over a number of years (see Hamermesh, 1993, Pencavel, 1986, Hart, 1987). A seminal article on the division of labour demand over employment and hours is Calmfors and Hoel (1988). In their model the firm chooses the number of workers and hours per worker to minimise labour costs, given the labour requirements for a fixed level of output, a fixed cost of employing each worker, the wage rate, standard hours and the overtime premium. This model yields the following well-known predictions: the demand for hours per worker is decreasing in standard hours, the wage rate and the overtime premium, and increasing in the fixed cost of labour. The negative relationship between standard hours and the demand for hours per worker, which may seem counterintuitive, derives from the fact that, when overtime is being worked, a reduction in standard hours increases the average labour cost per hour while the marginal cost remains constant. In the model of Calmfors and Hoel overtime hours are set by labour demand, but clearly actual overtime hours also involve supply considerations. The supply of overtime hours will respond to changes in the wage rate, depending on the preferences of individuals over income and leisure via the familiar income and substitution effects. While a higher wage induces substitution of work for leisure, a dominating income effect will reduce the hours supplied in favour of leisure (e.g. Nickell, 1983). An alternative to the labour demand or 'fixed-wage' approach above views hours worked as the outcome of a joint earnings-hours contract between firms and workers which reflects the preferences of both sides. Trejo (1991, 2003) postulates a compensating differential or 'fixed-job' model in which workers and firms agree on a package of weekly earnings and weekly hours. Within this fixed package, a change in standard hours or the overtime premium can be absorbed by an offsetting change in the (possibly implicit) straight-time hourly wage rate. As in the fixed-wage model, a reduction in standard hours yields

an increase in overtime hours. In further extensions, the presence of unions becomes an additional influence on overtime hours through bargaining over the conditions for overtime work (see, e.g., Contensou and Vranceanu, 2000).

Several recent studies report microeconometric evidence on the determinants of overtime work, mostly at the individual level (Trejo, 1993; Bell and Hart, 1998; Hunt, 1999; Bauer and Zimmermann, 1999). Hunt examines the effects of a reduction in standard hours on paid overtime for Germany. Using data from the German Socio-Economic Panel, she concludes that, at least for hourly paid workers in manufacturing, a one-hour reduction in standard working hours leads to a fall in actual hours of between 0.88 and 1.0, suggesting a small increase in overtime hours, of up to 0.12 hours. Bauer and Zimmermann, on the other hand, again for Germany, find that lower standard hours reduce overtime; this is in contrast to the predictions from the theoretical models. Bell and Hart, find that, for the UK, a rise in the basic hourly wage is associated with a fall in total weekly hours worked, consistent with the labour demand approach above. Trejo, on the other hand, indicates for the US that overtime hours increase with the wage rate, consistent with the substitution effect dominating in labour supply. On the role of union bargaining, Bell and Hart find no significant effect for the UK. For the US, on the other hand, Trejo concludes that the presence of unions increases the prevalence of premium pay for overtime, and reduces the incidence and hours of overtime.

The comparison of the results above is conditioned by the differing methodologies adopted in the various studies. Trejo (1993) uses a Tobit model estimated on a single cross section. The use of a Tobit model takes into account the censoring of overtime hours at zero (Tobin, 1958). However, a well-known limitation of the use of cross-section data is that one cannot control for correlations between the explanatory variables and unobserved individual specific effects such as preferences over income and leisure. If these unobserved characteristics are correlated with the explanatory variables parameter estimates are potentially biased. When available, panel data provide a route to overcoming this problem. Both Hunt (1999) and Bell and Hart (1998) exploit panel data to estimate a linear fixed effects model. While allowing for correlations between the explanatory variables and unobserved in these panel studies does not take into account the censoring of overtime hours at zero, and hence may yield inconsistent parameter estimates. Bauer and Zimmermann (1999) have panel data and take into account the censoring of overtime hours at zero but do not allow for correlations between the explanatory variables and unobserved individual specific effects. In this respect their approach is similar to the cross-sectional approach of Trejo (1993) discussed above.

The contribution of this paper is two-fold. On the methodological side to analyse overtime hours we use a panel data model that takes into account the censoring of overtime hours at zero for those not working overtime while also controlling for correlations between unobserved individual specific effects and the explanatory variables in a pre-specified way. This panel data model is based on Wooldridge (1995), Rochina-Barrachina (1999) and Kalwij (2003). To analyse the incidence of overtime we follow Hunt (1999). On the empirical side we extend the previous studies by analysing the effects on overtime incidence and hours of all three economic determinants identified above: changes in standard hours, the real wage rate, and unionisation. Furthermore, we examine the extent to which the estimates of the parameters of interest are biased when more restrictive econometric models are used, as in Trejo (1993), Bell and Hart (1998), Hunt (1999) and Bauer and Zimmermann (1999).

Our focus is the British economy. This provides a particularly interesting test-bed for the analysis of paid overtime work due to the absence of legal regulation of hours of work and overtime conditions. There is, for example, no parallel to the Standard Labor Act in the US, which mandates an overtime premium of 50% for hours worked beyond 40 per week. The recently introduced EU Working Time Directive may, at most, affect the last two years of our sample period. Traditionally unions have played an important role in negotiating the conditions for overtime working, such as the wage premium, as well as standard hours and wage rates. However, beginning with the Thatcher reforms, the last 20 years have seen deep-seated changes in relations between employers and employees, leading to the suggestion that the decline of union membership and collective agreements may have altered the use of overtime, potentially in the direction of employer preferences (Beatson, 1995).

For the estimation we exploit the New Earnings Survey Panel Dataset (NESPD). The NESPD provides a long and very large panel, covering about 165,000 employees per year over the period 1975-2001. This is the same dataset as used by Bell and Hart (1998, 1999a). In line with the studies discussed above, we analyse only paid overtime hours. The recent growth in unpaid overtime, notably by managers and professionals, is a separate phenomenon beyond the scope of this paper; an analysis is provided in Bell and Hart (1999).

The remainder of the paper is structured as follows: Section 2 describes the NESPD data and the construction of the sample, and gives some leading descriptive statistics. Section 3 presents the econometric framework. Section 4 reports the estimation results and comparisons with previous studies. Section 5 concludes.

Table I. Descriptive statistics

Year		1975	1980	1985	1990	1995	2000
Number of observat	ions	63245	91491	89750	98492	104850	102040
Female share in emp	loyment (%)	34.8	38.1	40.6	42.3	45.3	46.3
Panel A: Men							
Age (years)	average	39.5	39.7	39.0	38.7	38.7	39.8
Union coverage	%	57.9	48.4	48.0	34.8	26.7	17.6
Job tenure less than	one year, %	13.6	13.3	9.8	16.7	13.2	13.7
Basic pay per hour (a	<u>£</u> 2001)						
	average	7.6	7.8	8.6	9.7	10.6	11.4
	10th percentile	4.9	4.8	4.9	5.2	5.2	5.6
	50th percentile	6.9	7.0	7.6	8.4	9.0	9.6
	90th percentile	11.2	11.9	13.8	16.1	18.0	19.5
Standard working ho	ours (per week)						
	average	38.8	38.5	38.0	38.1	38.1	38.1
	10th percentile	36.0	35.0	35.0	35.0	35.0	35.0
	50th percentile	40.0	40.0	38.0	37.8	37.5	37.5
	90th percentile	40.0	40.0	40.0	40.0	40.0	41.5
Part-time work (less	than 30 standard hours) %	0.1	0.2	0.2	0.2	1.8	2.3
Overtime incidence	%	45.3	42.0	42.3	42.8	36.0	32.1
Overtime hours if w	orking overtime (per week)						
	average	8.5	8.7	8.2	8.6	8.3	7.I
	10th percentile	1.9	1.8	1.5	1.5	1.3	0.6
	50th percentile	7.4	7.5	7.0	7.1	6.5	5.4
	90th percentile	17.5	18.0	17.1	18.2	18.3	16.4

Year		1975	1980	1985	1990	1995	2000
Panel B: Women							
Age (years)	average	40.0	40.0	38.7	38.2	38.5	39.7
Union coverage	%	57.2	52.8	56.3	44.5	37.3	34.4
Job tenure less than o	one year, %	17.5	14.6	11.4	20.2	14.7	15.7
Basic pay per hour (#	2001)						
	average	5.3	5.4	5.9	7.0	7.9	8.5
	10th percentile	3.4	3.5	3.6	3.9	4 . I	4.4
	50th percentile	4.8	4.8	5.3	6.1	6.7	7.3
	90th percentile	7.7	8.0	9.2	11.5	13.4	14.6
Standard working ho	urs (per week)						
	average	32.4	31.2	31.0	31.4	31.0	30.6
	10th percentile	20.0	18.0	17.0	17.5	16.5	16.0
	50th percentile	36.0	35.0	35.0	35.0	35.0	35.0
	90th percentile	40.0	40.0	39.0	39.0	39.0	39.0
Part-time work (less	than 30 standard hours) %	29.9	34.7	32.6	30.9	32.8	36.0
Overtime incidence	%	10.2	10.1	15.9	18.9	19.8	17.6
Overtime hours if we	orking overtime (per week)						
	average	3.4	3.6	3.6	3.9	3.9	3.6
	10th percentile	0.5	0.5	0.5	0.6	0.6	0.3
	50th percentile	2.5	2.8	2.5	2.8	2.8	2.5
	90th percentile	7.5	8.0	8.0	8.8	9.0	8.4

Table I continued.

2 THE NEW EARNINGS SURVEY 1975-2001

The New Earnings Survey (NES) is an annual cross-section survey of earnings and hours of work at the individual level of employees in employment in Great Britain. The information is supplied by employees from payroll records and the data are gathered by the Office of National Statistics (ONS). All employees who have a National Insurance (NI) number that ends with a specified pair of digits form the sample. Since an NI number is issued to each individual at the minimum school-leaving age, the sampling frame for the Survey is a 1% random sample of the potential employed population. The same pair of terminating digits from the NI number is used as the basis for each year's sample, generating an unbalanced panel from the annual cross-section Surveys. The panel is characterised by new individuals entering the sample when starting employment and individuals exiting on retirement; in addition, unemployment and temporary withdrawal from the labour force generate gaps in individual participation. This unbalanced panel is referred to as the NESPD. In this study the panel covers the period 1975-2001.

The NESPD contains detailed information on the individual's gross weekly basic pay and overtime pay, standard working hours and paid overtime hours. Further information includes the employee's age, gender, job-tenure, union coverage and occupation, along with the employing establishment's industrial classification and regional location. As is most often the case with administrative data, the NESPD contains only limited information on individual characteristics and none at all on the household. In particular, human capital measures such as educational attainment, and household characteristics such as marital status and presence of children, are not available in the data.

2.1 DEFINITIONS AND DATA SELECTION

The dependent variable we use in the empirical analysis is weekly paid overtime hours, referred to as overtime hours. The Survey asks two questions on the individual's hours of work: 'Basic weekly hours of work. During the pay period, what was the average number of hours paid at basic rate of pay? (include all guaranteed hours, even if not worked)'; this is used as the measure of standard hours. And: 'Overtime hours during the pay period. How many overtime hours were worked during the pay period? (the hours must be consistent with the overtime earnings reported)'. The Survey does not distinguish different types of overtime hours, such as extra hours on the normal workweek, or additional weekend working. The overtime premium is not recorded, and no questions are asked on unpaid overtime. Where the

reported pay period is not a week data are converted by ONS to a weekly basis, located at the specified Survey week in April of each year.

The basic wage rate is calculated as basic pay divided by standard hours, where basic pay is defined in the Survey as gross weekly earnings, less overtime pay, bonus and incentive payments and any premium paid for shift-working. We convert all wage rates to 2001 £s, with deflation by the retail prices index. Responses to the question '*Has the employee worked at this job in your organisation for one year or more*?' allow us to determine whether the individual is a recent job-changer. Union coverage is based on whether the individual's pay and working conditions are set by a major industry or national collective agreement. The occupational classification is based on the 1990 Standard Occupational Classification (SOC) at the level of 9 Major Groups and the industrial classification on the 1995 Standard Industrial Classification (SIC) at the (alphabetical) one-digit level (13 groups). The region in which the worker's establishment is located is distinguished at the level of the 11 Standard Regions. Each of these classifications is confined to the fairly aggregated one-digit level in order to provide consistency across the observation period during which the classification systems were changed on several occasions.

Table AI in Appendix A provides a detailed overview of the data selection process. The raw unbalanced panel for 1975 to 2001 contains almost four and a half million observations on about half a million individuals. The reasons for dropping observations are presented incrementally, starting from hours of work as our central variable of interest. However, in many cases the same record is characterised by several items of deficient data, notably pay and hours jointly, so that the reason for discard may depend somewhat arbitrarily on sequence in the screening. Missing information on standard working hours or overtime hours leads to the exclusion of 16% of observations. Until 1994 an employee could be reported as working full-time but without specified hours. This was discontinued from 1995, leading to a fall in the number with missing hours from that date. Increasingly stringent requirements on contracts of employment over recent years have also led to more systematic specification of contractual hours of work. The fall in missing hours from 1998 is due to a change in the coding applied by ONS. Previously when pay was affected by absence hours were coded as missing; from 1998 they are set to standard hours, with these observations then removed due to missing pay information. Missing hours data prevent the calculation of the employee's hourly wage, leading to missing pay information as we require it. Records are also excluded on the grounds of missing pay information when earnings are not reported, pay is affected by absence, or the employee is a multiple jobholder with more than one employer. A small number of further cases are excluded because of missing values for the industry, occupation or regional codes (1.7%). All observations attributed to an individual are excluded if sex changes over the observation period (2%) or age does not increase by one year, plus or minus one, per

calendar year (age inconsistencies, 0.3%). In total, exclusions due to these data deficiencies cause a 22% reduction of the sample. In addition the Survey will capture individuals employed while in education or in retirement. To avoid results being influenced by this we exclude those aged under 21 or 60 and over (a further 10% reduction). To ensure robustness of the empirical results against outliers or extreme errors of measurement we exclude individuals with wage rates or hours in the top or bottom percentile of the respective distribution. This excludes on average 2.5% of the observations per year. Finally, the empirical analysis exploits the fact we have panel data and uses only those individuals who are observed in two consecutive periods. This causes an 8.4% reduction in the sample, with this selection of course largest in the end-years 1975 and 2001. The final sample used in the empirical analysis comprises 2,547,003 observations on 300,171 individuals with on average around 94,000 individuals present in any year.

2.2 PANEL ATTRITION

The NES is designed as a series of representative cross section Surveys, and it is the incidental feature that the same terminating digits on the NI number are used to draw the sample each year that makes it possible to construct the panel. Consequently, at the stage of data gathering little attention is paid to the panel aspect, and possible selection problems that may arise in the NESPD from sample attrition.

As mentioned above, the NES is framed as a 1% sample of the employed population. Table A2 in Appendix A shows that the actual percentage of the employed population covered by the NESPD in any year varies between 0.52 and 0.73%. Our final sample, after exclusions, is about one-third of one per cent of the employed population. Under-representation is to be expected given definitional differences, notably the exclusion from the NES of the self-employed (2.3 million individuals in 1975 rising to 3.3 million in 2001). Nevertheless, the non-response in the NESPD may be higher than one would expect in administrative data. As a check that coverage of female employment is not differentially restricted a comparison of the female share in employment between the OECD, based on the Labour Force Survey, and the NESPD shows that they are very close especially in the later years (Table A2, Appendix A). A comparison of working hours shows that the NESPD systematically under-samples smaller jobs (last two columns, Table A2). We return to this issue below.

Table A3, Appendix A, examines non-response from a further angle. The Table reports the percentage of individuals who are observed in the sample in both years t and t-s. In the final sample, 80% of men and 78% of women observed in year t were also present in year t-1, with 59% of men and 55% of women also observed in t-2. This percentage decreases towards 0% for years t and t-26. As expected

the percentages for women are rather lower than for men, reflecting their less systematic employment participation in the childcare years. In principle, if all individuals were in continuous employment from age 21 to age 59 then we should observe 100% for each value of s for the birth-cohorts 1942-1954. Table A3 shows that this is far from the case, and that for these cohorts the patterns are fairly similar to those of the sample of all birth-cohorts. We have to accept that non-negligible panel attrition is present in the NESPD despite the administrative nature of the data.

Absence from the NESPD while in employment can occur for two main reasons. The first relates to the way the data are gathered. Around 10% of the sample derives from responses provided directly to ONS by large public and private organizations from their computerised administrative databases. For the remaining 90%, employees with the appropriate NI numbers are located through the Inland Revenue register of current Pay-As-You-Earn (PAYE) income tax records, and a Survey questionnaire then sent to the employer for (compulsory) completion. The employer is not required to include in the tax register employees whose earnings fall below the tax threshold, potentially excluding many low-wage, short-hours workers. In 1999-2001, for instance, the threshold represented around 22 hours of work per week at the national minimum wage. However, in spite of the relatively high incidence of part-time work and low pay among women Table A2 does not support the inference that the procedure for sample location leads to the under-representation of women in the NESPD. Along with the presence of substantial numbers of low-wage, low-hours workers in the Survey this suggests that many employers find it convenient not to screen out those below the tax threshold from their tax returns, reducing attrition from this source. The second main reason for absence is the time lapse between the inspection of the tax register and the Survey week. This period between sampling and actual data gathering tends to be around one month. During this time a proportion of employees will have changed jobs and consequently no longer be on the payroll at the contacted employer. Thus employees with high rates of job turnover are likely to be under-represented in the sample. This kind of attrition would be endogenous in our study if a high rate of job turnover is associated with overtime hours. We examined this issue more formally by analysing the effects of overtime incidence and hours on the probability of being in the sample in the following year. The results of this analysis, available upon request from the authors, show that for almost all years there is no significant effect of overtime work on the probability of presence in the sample in the following year; the small number of significant cases includes both positive and negative effects. This suggests that endogenous selection is not a major issue for this study. Further, it can be argued that the unbalanced panel used here may suffer far less from endogenous selection than some of the data used in previous studies. The location of each year's sample independently through the current tax register means that an employee who was not in work or was

missed for any Survey should be located in any future employment. Attrition should not be cumulative. Having said this, we note that we do not have the information to control for endogenous selection if it were present. That sample attrition is exogenous is therefore a necessary assumption throughout the paper.

The advantages of the NESPD should also be emphasised. The consistency of the questionnaire over time and the ability to follow such a large sample of employees, on average 164.000 per year, for up to 27 years, is unique for micro-data. The questionnaire is sent to the employer who is under a legal obligation to complete it (Statistics of Trade Act 1947), ensuring a response rate significantly higher than in household based surveys. The data is provided from payroll records, giving a very high degree of accuracy in the hours and earnings information. Kalwij and Alessie (2003) report that the variance in earnings in the British Household Panel Study is four times that of the variance of earnings in the NESPD, underlining the much lower measurement error in administrative data.

2.3 DESCRIPTIVE STATISTICS

Table I reports the basic descriptive statistics for selected years, separately for men (Panel A) and women (Panel B). Over the sample period the British labour market underwent significant changes. The most notable features observable from the Table are the increase in the female share in total employment, from 35% in 1975 to 46% in 2000 (top of Table 1) and the decline in union coverage from 58% to 18% for men and from 57% to 34% for women. The average age of the workforce has remained fairly constant at around 40 years of age. The percentage of workers with job tenure of less than one year is highly cyclical, and rather higher for women than for men, consistent with their lower labour force attachment. The average real (basic) wage rate increased from \pounds 7.60 per hour to \pounds 11.40 for men and from \pounds 5.30 to \pounds 8.50 for women. Wage inequality widened, and at roughly the same rate among men and women. Median standard hours of work decreased from 40 to 37.5 for men and from 36 to 35 for work increased from 0.1% to 2.3% among men and from 30% to 36% among women. Appendix B reports the distribution of employment over occupations, industries and regions. In general employment has shifted from blue to white-collar occupations and from manufacturing, utilities and construction to service industries, notably financial and business services, and education.

Panel A shows that the incidence of paid overtime work for men remained over 42% until 1990 and declined thereafter to 32% in 2000. For women the pattern is reversed (Panel B): a sharp increase in the incidence of overtime from 10% in 1975 to 20% in 1990 was followed by only a marginal decrease to

around 18% in 2000. The average number of paid overtime hours worked by those working overtime fluctuates within relatively narrow bands, between 7.1 and 8.7 hours per week for men and between 3.4 and 3.9 hours for women. For men overtime hours can be observed to decrease in the second half of the 1990s. Appendix B shows that overtime incidence and overtime hours worked vary substantially across occupations and industries. For men high overtime jobs are in craft, semi-skilled and unskilled manual occupations, and in the transport, agriculture and engineering industries. For women they are in sales and machine operator jobs and again in transport industries. Overall Table 1 shows that in Britain paid overtime hours are a major and persistent feature of employment and not simply, or even primarily, a device for firms to accommodate cyclical variation in labour demand. This is in line with earlier observations for the UK during the 1950s and 1960s (Sallis, 1970) and for the US (Lilien and Hall, 1986).

3 ECONOMETRIC FRAMEWORK

This Section outlines the econometric framework for analysing the determinants of overtime incidence and of overtime hours for those working overtime. A central feature is allowing for correlations between the explanatory variables and the unobserved individual specific effects. The proposed framework is a reduced form approach, as in the studies discussed in the introduction. Ideally one would like to estimate a complete (structural) model explaining the changes in the incidence and hours of overtime simultaneously. There is, however, no standard econometric framework for estimating a panel data censored regression model in which one allows for (unrestricted) correlations between the observable characteristics and the unobserved individual specific effects. For an excellent survey of the estimation of nonlinear models with panel data we refer to Honoré (2002).

Section 3.1 follows Hunt (1999) and specifies a linear probability model of the determinants of the incidence of overtime. This model is estimated in first-differences to allow for possible correlations between the observable characteristics and the unobserved individual specific effects. Section 3.2 specifies a relatively new estimator for analysing overtime hours for those who work overtime, based on Rochina-Barrachina (1999) and Kalwij (2003). These authors propose an estimation strategy of a censored regression model that eliminates the individual specific effects from the main equation by taking first-differences. Both studies show that taking first-differences significantly reduces the sensitivity of the estimates to a specific parameterization of the individual specific effects based on the so-called Mean Conditional Independence assumption (Wooldridge, 1995). The proposed model in this study is estimated using a two-step estimator in the spirit of Heckman (1976). This model nests the model estimated by Hunt (1999) for analysing overtime hours conditional on working overtime.

Throughout this section the basic notation is as follows: the index for the individual is *i*, the number of individuals is equal to N, and the index for calendar year *t*. Overtime incidence is denoted by I_{it} (a dummy variable), overtime hours is denoted by V_{it} , and the explanatory variables are denoted by X_{it} .

3.1 OVERTIME INCIDENCE

We follow Hunt (1999) and specify a linear probability model to analyse the incidence of overtime. Overtime incidence I_{it} is assumed to relate to the explanatory variables X_{it} as follows:

(1)
$$I_{it} = X_{it}\delta + \zeta_i + \eta_{it}.$$

The error term η_{it} is an idiosyncratic term with mean zero and assumed independent of the explanatory variables. The unobserved individual specific effect is denoted by ζ_i and is allowed to correlate with the explanatory variables. The advantage of using a linear probability over, for instance, a probit model, is that the unobserved individual effect can be differenced out:

(2)
$$\Delta I_{it} = \Delta X_{it} \delta + \Delta \eta_{it},$$

where $\Delta I_{ii} = I_{ii} - I_{ii-1}$, $\Delta X_{ii} = X_{ii} - X_{ii-1}$ and $\Delta \eta_{ii} = \eta_{ii} - \eta_{ii-1}$. Equation (2) is estimated using a Least Squares estimator and the standard errors are adjusted for heteroscedasticity and clustering, since we have multiple observations per individual. An empirical consequence of employing a first-difference estimator is that age and year effects are no longer separately identified (Heckman and Robb, 1985). Therefore we cannot report on the parameter estimates of age and we omit the estimates of the time-dummy variables from the results Tables since they include age effects. We return to this issue in Section 4.3.

3.2 OVERTIME HOURS

We employ a relatively new econometric method for analysing overtime hours. This estimator is based on Tobin (1958), Wooldridge (1995), Rochina-Barrachina (1999) and Kalwij (2003). Overtime Hours V_{it} is assumed to relate to the explanatory variables X_{it} as follows:

(3)
$$V_{it}^* = X_{it}\beta + \alpha_i + \varepsilon_{it},$$

(4)
$$V_{it} = \max(0, V_{it}^*),$$

where V_{it}^* is the latent dependent variable and V_{it} is the observed overtime hours and is censored at zero. The unobserved individual specific effect is denoted by α_i and is allowed to correlate with the explanatory variables. The error term ε_{it} is assumed to be a normal random variable with mean zero and variance σ_t^2 , $\varepsilon_{it} \sim N(0, \sigma_t^2)$, and is allowed to be serially correlated.

Following Kalwij (2003) the first step is to eliminate the individual specific effects from the main equation, equation (3), by taking first differences:

(5)
$$\Delta V_{it}^* = \Delta X_{it} \beta + \Delta \varepsilon_{it}$$

(6)
$$\Delta V_{it} = \begin{cases} \Delta V_{it}^* & if \quad V_{it-1}^* > 0 \quad and \quad V_{it}^* > 0 \\ 0 & if \quad V_{it-1}^* \le 0 \quad and \quad V_{it}^* \le 0 \\ V_{it}^* & if \quad V_{it-1}^* \le 0 \quad and \quad V_{it}^* > 0 \\ -V_{it-1}^* & if \quad V_{it-1}^* > 0 \quad and \quad V_{it}^* \le 0 \end{cases}$$

where $\Delta V_{it}^* = V_{it}^* - V_{it-1}^*$, $\Delta V_{it} = V_{it} - V_{it-1}$ and $\Delta \varepsilon_{it} = \varepsilon_{it} - \varepsilon_{it-1}$. Although the individual specific effects are eliminated from Equation (5), estimating it by Least Squares on a sub-sample of individuals with positive hours of overtime in periods t and t-1 yields inconsistent estimates because, generally, $E[\Delta \varepsilon_{it} | V_{it-1}^* > 0, V_{it}^* > 0] \neq 0$. Therefore a correction for censoring has to be applied that is somewhat more complicated than in the cross-sectional case (Tobin, 1958). A further complication is that, as implied by equation (6), the individual specific effects will still appear in the correction term for the censoring of overtime hours at zero (Kalwij, 2003).

To deal with possible correlations between the unobserved individual specific effects and the explanatory variables we make the so-called Mean Conditional Independence assumption (Wooldridge, 1995). The Mean Conditional Independence assumption made in this study is:

(7)
$$\alpha_i = \overline{X}_i \gamma + \mu_i,$$

where \overline{X}_i is the average over time of the explanatory variables and the random individual specific effect μ_i is assumed to be a normal random variable with mean zero and variance σ_{μ}^2 , $\mu_i \sim N(0, \sigma_{\mu}^2)$. This specification is a popular one because in the case of a linear regression model this would yield the familiar fixed effects estimates (Mundlak, 1978). The Mean Conditional Independence assumption is implemented empirically by substituting equation (7) in equation (3):

(8)
$$V_{it}^* = X_{it}\beta + \overline{X}_i\gamma + u_{it},$$

(9)
$$V_{it} = \max(0, V_{it}^*),$$

with $u_{it} = \mu_i + \varepsilon_{it}$ and $\sigma_{u,t}^2 = \sigma_{\mu}^2 + \sigma_t^2$. The correlation between u_{it-1} and u_{it} is denoted by ρ_t . Kalwij (2003) derives a Maximum Likelihood estimator for a model as specified by equations (8) and (9) based on taking first differences of the equation of interest, as in equations (5) and (6), thereby eliminating the individual specific effect from the main equation. He further shows that taking first-differences helps to alleviate the sensitivity of the estimates to a specific parameterization of the individual specific effects (equation (7)). A more convenient way of estimating this proposed first-difference panel data Tobit model is to use a two-step estimator, in the spirit of Heckman (1976). A two-step estimator for a panel

data selection model based on taking first differences of the equation of interest is developed by Rochina-Barrachina (1999). Kalwij (2003a) presents a formal derivation and shows that a two-step firstdifference estimator for a panel data Tobit model is a special case of her estimator. We develop this as the econometric framework for analysing overtime hours.

The expectation of the first-difference of equation (8) conditional on the individual being observed to work overtime in two consecutive years is given by (see Kalwij, 2003a):

(10)
$$E[\Delta V_{it} | V_{it-1}^* > 0, V_{it}^* > 0] = \Delta X_{it} \beta + \pi_t \Lambda_t (M_{it-1}, M_{it}, \rho_t) - \pi_{t-1} \Lambda_{t-1} (M_{it-1}, M_{it}, \rho_t)$$

with

$$\begin{aligned} \pi_{t-1} &= (\sigma_{t-1} - \rho_t \sigma_t), \\ \pi_t &= (\sigma_t - \rho_t \sigma_{t-1}), \\ M_{it-1} &= (X_{it-1}\beta + \overline{X}_i\gamma) / \sigma_{u,t-1}, \\ M_{it} &= (X_{it}\beta + \overline{X}_i\gamma) / \sigma_{u,t}, \\ \Lambda_{t-1}(M_{it-1}, M_{it}, \rho_t) &= \frac{\phi(M_{it-1}) \Phi((M_{it} - \rho_t M_{it-1}) / \sqrt{1 - \rho_t^2})}{\Phi^2(M_{it-1}, M_{it}, \rho_t)} \\ \text{and} \qquad \Lambda_t (M_{it-1}, M_{it}, \rho_t) &= \frac{\phi(M_{it}) \Phi((M_{it-1} - \rho_t M_{it}) / \sqrt{1 - \rho_t^2})}{\Phi^2(M_{it-1}, M_{it}, \rho_t)}. \end{aligned}$$

The standard normal distribution function is denoted by $\Phi(.)$, the standard normal density function by $\phi(.)$ and the bivariate standard normal distribution function is denoted by $\Phi^2(.)$. The two correction terms on the RHS of equation (10) are now added and subtracted from the first-difference of equation (8):

(11)
$$\Delta V_{it} = \Delta X_{it} \beta - \pi_t \Lambda_t (M_{it-1}, M_{it}, \rho_t) + \pi_{t-1} \Lambda_{t-1} (M_{it-1}, M_{it}, \rho_t) + \xi_{it},$$

where ξ_{ii} is an error term with a conditional expectation equal to zero, i.e. conditional on the individual being observed to work overtime in two consecutive years.

Following Kalwij (2003a), equation (11) is estimated in two steps. In the first step the following bivariate probit model is estimated:

$$V_{it-1}^* = X_{it-1}\beta + \overline{X}_i\gamma + u_{it-1} \qquad V_{it}^* = X_{it}\beta + \overline{X}_i\gamma + u_{it},$$

$$I_{it-1} = \begin{cases} 0 & if \quad V_{it-1}^* \le 0 \\ 1 & if \quad V_{it-1}^* > 0 \end{cases}, \qquad I_{it} = \begin{cases} 0 & if \quad V_{it}^* \le 0 \\ 1 & if \quad V_{it}^* > 0 \end{cases}$$

The dependent variables are the indicators for working overtime in the two periods, i.e. I_{it-1} and I_{it} .

Using Maximum Likelihood to estimate this bivariate probit model yields estimates for $\beta_{t-1} = \frac{\beta}{\sigma_{u,t-1}}$,

 $\beta_t = \frac{\beta}{\sigma_{u,t}}, \ \gamma_{t-1} = \frac{\gamma}{\sigma_{u,t-1}}, \ \gamma_t = \frac{\gamma}{\sigma_{u,t}}, \ \text{and} \ \rho_t.$ These first step estimates are obtained for each pair of

consecutive years separately. Given that we have 27 years of data (see Section 2) this means we estimate 26 bivariate probit models.

The estimates $\hat{\beta}_{t-1}$, $\hat{\beta}_t$, $\hat{\gamma}_{t-1}$ and $\hat{\gamma}_t$ are used to obtain $\hat{M}_{it-1} = X_{it-1}\hat{\beta}_{t-1} + \overline{X}_i\hat{\gamma}_{t-1}$ and $\hat{M}_{it} = X_{it}\hat{\beta}_t + \overline{X}_i\hat{\gamma}_t$. Together with the estimate $\hat{\rho}_t$ we obtain estimates of the correction terms. These are substituted in equation (11):

(12)
$$\Delta V_{it} = \Delta X_{it} \beta - \pi_t \Lambda_t (\hat{M}_{it-1}, \hat{M}_{it}, \hat{\rho}_t) + \pi_{t-1} \Lambda_{t-1} (\hat{M}_{it-1}, \hat{M}_{it}, \hat{\rho}_t) + \xi_{it}.$$

In the second step equation (12) is estimated using a Least Squares estimator on the sub-sample of all pairs of observations for which $V_{it-1} > 0$ and $V_{it} > 0$, i.e. where we observe positive values of overtime hours in two consecutive years. Clustering is taken into account when calculating the standard errors. This two-step procedure yields consistent estimates of β . The parameters π_{t-1} and π_t are treated as auxiliary parameters. In our empirical analysis we have 27 years of data, which implies we have 2x26 additional selection terms (t = 1976-2001).

Applied to the analysis of overtime, the econometric model of equation (12) extends the models chosen by Bell and Hart (1998) and Hunt (1999) by controlling for the censoring of overtime hours at zero. Both Bell and Hart (1998) and Hunt (1999) impose on equation (12) the restriction that π_{t-1} and π_t are each equal to zero. We test these restrictions to assess whether or not taking the censoring into account is of empirical importance. In addition Bell and Hart (1998) in essence estimate equation (12) using all observations, i.e. including those with zero overtime hours.

4 EMPIRICAL RESULTS

In the results reported in this Section we analyse the extent to which overtime work (incidence and hours) is affected by observed characteristics. As explanatory variables we include standard working hours, the basic wage rate, union coverage, recent job change (tenure less than one year), occupation, and the industry and regional location of the firm where the individual is employed. For women we also include an indicator of whether she works part-time. Sections 4.1 and 4.2 discuss the estimation results and perform sensitivity analyses for overtime incidence and hours respectively. Section 4.3 examines the extent to which changes in the observable characteristics explain the actual changes observed over the period 1975-2001.

4.1 OVERTIME INCIDENCE

Table 2 reports the results from estimation of the first-stage model of overtime incidence, as outlined in Section 3.1. The estimates are presented as the marginal effects on the probability of working overtime.

Table 2 shows that a reduction of one hour in standard hours increases the probability of working paid overtime by 0.0067 for men and 0.0033 for women. This is significantly lower than the effects predicted by Hunt (1999) using German data: 0.029 for men working in manufacturing & construction and 0.017 for men working in services. Hunt's model does not include any explanatory variables apart from year dummies. When we replicate the specification in Hunt (1999) we still find significantly lower effects: a one-hour reduction in standard hours reduces overtime incidence by 0.0073 for men working in manufacturing & construction and by 0.0036 for men working in services (full set of results not reported here). For women, after controlling for standard hours, the effect of working part-time is insignificant. The wage rate has a significant negative effect on the incidence of overtime for both men and women: a one-percent increase in the basic wage rate decreases the incidence of overtime by 0.068% for men and 0.0103% for women. Union coverage increases the incidence of overtime for men by 0.0082, but has an insignificant effect for women. This conclusion contrasts, in both magnitude and direction, with the finding by Trejo (1993) that, for hourly paid workers in the US, unionisation (defined as covered workers plus union members) reduces overtime incidence by 17%. However, when Trejo adds a control for the probability that an overtime premium is paid, the effect of union coverage becomes insignificant and more in line with our results. This may indicate that unobservable characteristics do matter and failure to control for them may affect the parameter estimates. We return later to this issue. A further possibility is that the NESPD question, restricting coverage to specified major national or industry level

agreements, is a weak measure of the role of union coverage. Recent job-change has no effect on overtime incidence for men, but for women it lowers the probability of working overtime by 0.0024. As mentioned in Section 3, age effects are not identified.

Table 2. Estimation results: equation (2).	Time effects are included.	The dependent variable is the	change in the
incidence of overtime.			

larginal effects on overtime incidence Men		Women		
Explanatory variables	p.e.	s.e.	p.e.	s.e.
Part-time			-0.0010	0.0033
Standard hours	-0.0067	0.0003	-0.0033	0.0002
Log-hourly basic pay	-0.0680	0.0034	-0.0103	0.0033
Union coverage	0.0082	0.0019	-0.0002	0.0020
Job tenure less than one year	0.0004	0.0012	-0.0024	0.0011
Age squared	-1.8E-04	I.3E-05	-8.4E-06	I.2E-05
Occupation				
Managers and administrators	-0.1226	0.0064	-0.0583	0.0101
Professional	-0.0906	0.0067	-0.0424	0.0103
Associate professional/ technical	-0.0881	0.0070	-0.0558	0.0100
Clerical and secretarial	-0.0648	0.0066	-0.0416	0.0092
Craft and related occupations	-0.0216	0.0058	-0.0225	0.0114
Personal and protective services	-0.0199	0.0102	-0.0486	0.0099
Sales	-0.1332	0.0077	-0.0236	0.0100
Plant and machine operatives	0	-	0	-
Other	-0.0186	0.0076	-0.0554	0.0099
Industry				
Agriculture	-0.0525	0.0218	0.0088	0.0322
Energy and water supply	-0.0253	0.0138	0.0530	0.0181
Mineral extractions, metal and chemical				
products	0.0033	0.0093	0.0062	0.0114
Metal goods and engineering products	0.0074	0.0076	0.0296	0.0089
Other manufacturing industries	0	-	0	-
Construction	-0.0065	0.0096	-0.0081	0.0118
Distributive trades and catering	-0.0270	0.0073	0.0173	0.0071
Transport and communications	0.0484	0.0099	0.0406	0.0118
Financial services	-0.0281	0.0079	0.0116	0.0073
Public administration	-0.0438	0.0102	-0.0054	0.0085
Educational services	-0.0822	0.0128	-0.0273	0.0084
Health services	-0.0175	0.0167	-0.0094	0.0088
Business and other services	-0.0402	0.0100	0.0018	0.0080

Region				
Greater London	0	-	0	-
South-East (excluding Greater London)	0.0032	0.0062	0.0157	0.0070
East Anglia	-0.0059	0.0131	0.0179	0.0153
South West	-0.0046	0.0102	0.0497	0.0121
West Midlands	-0.0031	0.0103	0.0116	0.0125
East Midlands	0.0053	0.0111	-0.0073	0.0134
Yorkshire & Humberside	-0.0111	0.0108	0.0060	0.0128
North West	-0.0152	0.0107	-0.0036	0.0123
North	0.0077	0.0166	0.0299	0.0165
Wales	-0.0017	0.0163	0.0245	0.0177
Scotland	0.0195	0.0144	0.0116	0.0159
N	1177724		826702	
R ²	0.0044		0.0024	

Table 2. Continued

The role of occupation as a determinant of overtime working is clearly established in Table 2. Plant and machine operatives (semi-skilled manual workers – the reference category) are more likely than other occupations to work overtime. Blue collar workers are more likely to work overtime than white collar workers, with managers and professionals the least likely (although, of course, the latter group are most likely to work unpaid overtime - see Section 1). For women the differences across occupations are still significant but smaller. Estimates of the industry effects show that especially employees in the transport sector, both men and women, have a relatively high incidence of overtime, while the education sector makes conspicuously little use of paid overtime. Regional differences are jointly insignificant for men and only marginally significant for women.

A further empirical issue with a linear probability model is that the predicted individual probabilities may lie outside the zero-one interval in small samples. On checking, our relatively large sample did not reveal any unusually large predicted changes in the individual probabilities of working overtime that would cause the probability to lie outside the zero-one interval.

Previous studies, with the exception of Hunt (1999) as discussed above, use econometric models that do not control for unobserved individual specific effects that are allowed to correlate with the explanatory variables. Most often this is due to data limitations. As a sensitivity analysis on this aspect of specification Table 3 reports estimates using a random effects estimator, as employed by Bauer and Zimmermann (1999) and in essence also Trejo (1993) since he estimates equation (1) on a single cross-section. A formal test of a random versus a fixed effects model, or rather a general model specification test, is the well-known Hausman test (Hausman, 1978). The Hausman test statistics reported at the

bottom of Table 3 show that the random effects specification is clearly rejected by the data. This indicates that estimates using a random effects estimator will be biased and any inferences based on a cross-section model or random effects panel data model have to be made with extreme caution. The estimates in Table 3 can be compared with those at the top of Table 2 to give insights into the empirical importance of controlling for unobserved individual specific effects that are allowed to correlate with the explanatory variables. For men and women, the random effects specification biases the estimated effect of standard hours on overtime incidence upwards (i.e. smaller negative). This may explain why Bauer and Zimmermann (1999), for Germany, find that lower standard hours reduce overtime. For men, the random effects specification overestimates the effect of basic pay on overtime incidence by a factor of two; for women, the absolute bias is smaller but switches sign. The effect of age on overtime incidence is identified because orthogonality is assumed between age and the unobserved individual specific effect; implicitly, also, there are no birth-cohort effects. Clearly, since this model is rejected one cannot interpret the estimates of the age effects. The overall conclusion is that using a random effects specification yields severely biased parameter estimates.

Table 3. Sensitivity analysis: estimation of	equation (1) with	μ_i treated as a random	individual specific	effect (Ba	uer and
Zimmermann, 1999).*					

Dependent variable: overtime incidence	Men		Women	
	p.e.	s.e.	p.e.	s.e.
Age	0.0135	0.0005	0.0063	0.0024
Part-time			-0.0009	0.0004
Standard hours	-0.0041	0.0002	-0.0007	0.0001
Log-hourly basic pay	-0.1495	0.0023	0.0086	0.0020
Union coverage	0.0315	0.0019	-0.0045	0.0015
Job tenure less than one year	0.0016	0.0014	0.0043	0.0013
Age squared	-1.7E-04	6.5E-06	-4.4E-06	5.0E-06
Ν	1177724		826702	
R ²	0.1629		0.0362	
Hausman model missspecification test.				
(Null-hypothesis: no misspecification)	88.2 ª		l 9.3⁵	

* Controls are also included for the year of observation, the individual's occupation, and the industry and region of the firm at which employed.

^a Critical Value is $F(36, \infty) = 1.93$.

^b Critical Value is $F(37, \infty) = 1.91$.

4.2 OVERTIME HOURS

The results are reported in Table 4 for the second stage of the model as outlined in Section 3.2. The estimates are interpreted as the marginal effects on overtime hours when working overtime. At the bottom of Table 4 we report on a joint test of the significance of the parameters corresponding to the correction terms for positive hours of overtime in equation (12). In the equation for men we reject the null-hypothesis that all the π_{t-1} 's and π_t 's are equal to zero, although we do not reject it for women. We return to this when discussing Table 5 below.

Table 4 shows that a one-hour reduction in standard hours increases overtime hours by 0.13 hours for men and 0.16 hours for women, both significant. This negative effect is in line with theoretical predictions and with the effect on overtime incidence. Hunt (1999), however, estimates an insignificant effect for men working in manufacturing & construction and a 0.38 hours increase for men working in services. Using exactly the same specification as Hunt we still find significantly different effects: a one-hour reduction in standard hours brings a 0.20 hours increase in overtime for men employed in manufacturing & construction and 0.15 hours increase for men in services. Bell and Hart (1999a), using cross section NES data for men in 1996, report that a one-hour reduction in standard hours results in a 0.087 increase in overtime hours, about two-thirds of our estimate. The major departure both from theoretical predictions and from our findings on overtime incidence is on the effect of the hourly wage. A 1% increase in the basic wage rate increases overtime hours by 0.0105 hours for men and 0.0017 hours for women (Table 4). In contrast to our findings Bell and Hart (1998, 1999a) report a (small) decrease in overtime hours. For the US, Trejo (1993) also reports a positive effect. We find that union coverage does not significantly reduce overtime hours; this is in line with Bell and Hart (1998, p.346) and Trejo (1993, Table 9). A recent job change does not significantly affect overtime hours.

Table 4. Estimation results: equation (12). Time effects are included. The dependent variable is the change in weekly paid overtime hours

Effect on Overtime Hours		Men		Women
Explanatory variables	p.e.	s.e. s.	p.e.	s.e.
Part-time			-0.1630	0.1662
Standard hours	-0.1305	0.0110	-0.1553	0.0108
Log-hourly basic pay	1.0534	0.1138	0.1714	0.1597
Union coverage	-0.0859	0.0464	0.1052	0.0692
Job tenure less than one year	0.0411	0.0319	0.0670	0.0363
Age squared	-0.0013	0.0004	-0.0010	0.0006
Occupation				
Managers and administrators	0.2909	0.2023	-1.2319	0.3379
Professional	-0.0844	0.1917	-1.1612	0.3348
Associate professional/ technical	-0.2639	0.1938	-1.4791	0.3655
Clerical and secretarial	-0.2987	0.1526	-1.3651	0.3035
Craft and related occupations	-0.1130	0.1255	-0.3376	0.3585
Personal and protective services	0.0435	0.2764	-0.5692	0.4059
Sales	0.4237	0.2647	-1.3703	0.3595
Plant and machine operatives	0	-	0	-
Other	0.0513	0.1638	-0.6247	0.4122
Industry				
Agriculture	-0.3942	0.5836	0.0168	0.9727
Energy and water supply	-0.6269	0.4322	0.2630	0.7232
Mineral extractions, metal and chemical products	-0.7342	0.2681	0.5121	0.6857
Metal goods and engineering products	-0.8878	0.2068	0.2321	0.4018
Other manufacturing industries	0	-	0	-
Construction	-0.5686	0.2652	0.2483	0.7304
Distributive trades and catering	-0.9171	0.2284	-0.2704	0.3756
Transport and communications	0.2251	0.2637	0.0785	0.4965
Financial services	-0.5391	0.3033	-0.4557	0.3803
Public administration	-0.8392	0.3364	0.1377	0.5086
Educational services	0.4137	0.5756	0.5419	0.5828
Health services	-1.8958	0.6685	-0.3728	0.5662
Business and other services	-0.3445	0.3623	0.0258	0.4525

Region				
Greater London	0	-	0	-
South-East (excluding Greater London)	-0.2059	0.2792	-0.2558	0.2742
East Anglia	-0.3961	0.5545	-0.0191	0.5961
South West	-0.9085	0.4219	0.4074	0.4953
West Midlands	-0.2865	0.4437	0.0452	0.4951
East Midlands	-0.5221	0.4387	0.3820	0.5628
Yorkshire & Humberside	-0.5069	0.4537	0.5372	0.4284
North West	-0.4390	0.4571	0.1017	0.5320
North	0.8069	0.6250	0.0422	0.8575
Wales	0.2985	0.6266	0.9273	0.6171
Scotland	-0.2800	0.5744	0.4068	0.6419
N	24/000		70077	
N	346909		/20//	
R ²	0.0083		0.0219	
F-test on the sample correction terms	6.97 ^a		1.37ª	

Table 4. Continued

^a Critical Value is $F(52,\infty)=1.89$.

For both men and women the estimates of occupation, industry and region effects are often insignificant. Occupational differences are jointly insignificant for men but jointly significant for women. Industry differences are jointly significant for men but jointly insignificant for women. Regional differences are jointly insignificant for both men and women. The main significant differences are: men in Health services and in Distributive trades work relatively fewer overtime hours, and women working as Managers and Professionals work relatively fewer overtime hours.

Direct comparison of our results with those obtained in other studies is limited by the different econometric methods used. To gain some insights into this issue we report in Table 5 the results we would obtain applying the differing econometric methodologies suggested in Trejo (1993), Bell & Hart (1998, 1999a), Hunt (1999) and Bauer and Zimmermann (1999). We cannot adopt their models exactly, and there are some minor differences such as our use of a first-difference estimator instead of a fixed effects 'within' estimator. But the underlying assumptions we make in estimating these alternative models are the same as in the originals, or even slightly less stringent, making the comparison favourable

to their alternative approaches. In spite of these caveats, the exercise is extremely useful from a methodological point of view. The results of Table 5 compare with those at the top of Table 4.

	E . 1.67					
	First-difference		First-difference		Random Effects	
Cells: Marginal Effects	all ob	oservations if working overtime		Tobit		
						Trejo (1993)
	Bell & H	art (1998)			Bella	& Hart (1999)
	Hu	unt (1 999)	Hur	nt (1999)	Bauer & Zimme	rmann (1999)
Men	p.e.	s.e.	p.e. s.e.		p.e.	s.e.
Age					0.2909	0.0131
Standard hours	-0.1196	0.0039	-0.1660	0.0104	-0.0245	0.0083
Log-hourly basic pay	-0.5882	0.0474	0.4400	0.1016	-1.3564	0.1017
Union coverage	0.0944	0.0233	0.0154	0.0452	-0.2856	0.0436
Job tenure less than one year	0.0292	0.0133	0.0224	0.0316	0.2026	0.0408
Age squared	-0.0024	0.0001	-0.0024	0.0004	-0.0034	0.0002
Ν	1177724		346909		346909	
R ²	0.0061		0.0072		0.0760	
Hausman model missspecication test.						
Null-hypothesis is no misspecification					25.70 ^a	
Women	p.e.	s.e.	p.e.	s.e.	p.e.	s.e.
Age					0.0588	0.0128
Part-time	-0.0338	0.0242	-0.1671	0.1660	0.7113	0.1036
Standard hours	-0.0422	0.0017	-0.1570	0.0107	-0.0342	0.0058
Log-hourly basic pay	-0.0881	0.0266	0.2026	0.1591	0.1431	0.0725
Union coverage	-0.0010	0.0118	0.1009	0.0661	-0.1001	0.0495
Job tenure less than one year	0.0007	0.0067	0.0671	0.0360	0.1260	0.0452
Age squared	-0.0001	0.0001	-0.0008	0.0005	-0.0008	0.0002
Ν	826702		72077		72077	
R ²	0.005		0.0208		0.0962	

Table 5. Sensitivity analysis: comparison with estimates obtained using the methodology of previous studies. The dependent variable is weekly paid overtime hours for those working overtime.

Hausman model missspecification	
test.	
Null-hypothesis is no misspecification	4.48

* Controls are also included for the year of observation, the individual's occupation, and the industry and region of the firm at which employed. a Critical Value is $F(36, \infty) = 1.93$.

b Critical Value is $F(37, \infty)=1.91$.

Bell and Hart (1998) and Hunt (1999, Table I, first columns) essentially estimate the main equation in first differences using all observations, setting overtime hours to zero for those who do not work overtime. This is the first column of results in Table 5. The effect of standard hours does not change much for men but is substantially overestimated (i.e. smaller negative) for women (-0.042 versus – 0.155). The effects of basic pay are severely biased and even switch signs. The same holds for union coverage. Overall we conclude that the approach of taking first-differences of overtime hours (or, by extension, total hours) without controlling for the censoring of overtime hours at zero yields severely biased estimates.

Column 2, Table 5, reports the estimation results when we take first-differences of overtime hours for those who work overtime in two consecutive years, but without controlling for sample selection, i.e. equation (12) with $\pi_{t-1} = \pi_t = 0$. Hunt (1999, Table I, last column) also estimates this model. As discussed above, for women taking into account censoring of overtime hours at zero turned out to be of no empirical importance when estimating the model in first-differences. Hence it is not surprising that we find that for women the results in Table 4 and Table 5, second column, are not significantly different from each other. For men the results are significantly different; in particular, the effect of the wage rate is severely under-estimated. Hence, we conclude that controlling for the censoring of overtime hours at zero is of empirical importance.

The third column reports on random effects Tobit estimates, hence assuming there are no correlations between the unobserved individual specific effects and the explanatory variables. Here we use a twostep estimator (Heckman, 1976). The Hausman test statistics are reported in the bottom rows. For both men and women we reject the random effects model. When comparing the estimates of Table 5, third column, with those in Table 4 one has to conclude that estimates of the parameters of interest in this study based on a single cross-section or a random effects specification when using panel data are severely biased.

4.3 PREDICTIONS

The results above suggest that most of the included explanatory variables have significant but small effects on overtime work. Table 6 quantifies their estimated impact on changes in overtime incidence and hours for those working overtime over the observation period 1975-2001. For these within sample predictions the observed changes in the explanatory variables (in parentheses, Table 6) are used together with the parameter estimates of Tables 2 and 4.

Table 6. The predicted effects on overtime incidence and hours of the observed change in characteristics, 1975-2001.

		Predicted effect on
Period 1975-2000	Predicted effect on	overtime hours conditional on
	overtime incidence	working overtime
Characteristics	%-points	Hours
Panel A: Men		
Standard hours (-0.8 hours per week)	0.51	0.10
Hourly basic pay (+50%)	-3.43	0.53
Union coverage (-40%-points)	-0.33	0.03
Job tenure less than one year (0.1%-points)	0.00	0.00
Occupational structure	-1.53	0.04
Industrial structure	-0.92	0.02
Regional structure	-0.01	-0.02
Total explained	-5.71	0.70
Total change	-13.23	-1.42

Panel B: Women

Part-time (+6.1%-points)	-0.01	-0.01
Standard hours (-1.8 hours per week)	0.60	0.28
Hourly basic pay (+62%)	-0.64	0.11
Union coverage (–23%)	0.01	-0.02
Job tenure less than one year (-1.8%-points)	0.00	0.00
Occupational structure	-0.40	-0.12
Industrial structure	-0.22	-0.06
Regional structure	0.14	-0.01
Total explained	-0.52	0.17
Total change	7.43	0.21

Table 6, Panel A, shows that for men the model can explain 43% of the observed change in overtime incidence from 1975-2001. The actual decrease in standard hours of 0.8 hours resulted in an increase in the incidence of overtime of 0.51 percentage points. The biggest contributions to the observed decline in overtime incidence came from basic pay, where the 50% increase over the period brought a reduction of 3.43 %-points, and changes in the occupational and industry structure of employment, bringing a further decrease of 2.45 %-points. For women the model is much less successful, predicting a small decrease of 0.52% in overtime incidence, against the actual increase of 7.4%. The rise in basic pay and the changing occupational and industry structure of employment all predict a fall in the incidence for women effectively remains unexplained. For overtime hours the relative predictive performance for men and women is reversed. For women the model predicts an increase in overtime hours worked of 0.17, against an actual increase of 0.21, with the largest contribution again coming from the reduction in standard hours. For men, on the other hand, the model predicts a small increase of 0.70 in overtime hours, with almost all the variables identified contributing to an increase, while the recorded change was a decline of 1.42 hours.

These results suggest that there are further influences on both overtime incidence and hours not captured in our model. In the case of overtime working by women we have not been able to control for

employment participation. As discussed in Section 2, being employed is a condition for being in the sample. Since the employment decision will be influenced by the wage, the estimated coefficients are likely to be biased. Economic analyses point to the overtime premium and fixed costs of employment (such as personnel costs and training expenditures) as further determinants of overtime work, neither of which are available in our data. We also note the changes in contractual work patterns that are occurring, such as the shift to more flexible hours across workweeks, including the use of annualised hours. Where these are not constant for the individual worker in consecutive periods and are correlated with the included variables omitted variable bias will occur. So while we are confident that our estimates improve on earlier studies by reducing the various biases we have identified, other potential sources of bias remain to be addressed if suitable data become available.

5 CONCLUSIONS

This paper has investigated the use of paid overtime work in Britain using individual-level data from the New Earnings Survey from 1975-2001. A panel data model has been employed allowing for censoring of paid overtime hours at zero and for correlations between the explanatory variables and unobserved individual specific effects. In particular we examined the effects of standard working hours, the basic wage rate and unionisation, as well as the occupational and industrial structure of employment, on paid overtime incidence and on hours conditioned on working overtime. The main empirical findings are summarized as follows:

Over this 27-year period paid overtime working was a major and persistent feature of employment. Over 40% of men were working paid overtime in the Survey week each year until 1990, and in 2000 the proportion was still 32%. Among women, the incidence was lower, but rising quite sharply, from 10% to around 18%. For those working overtime the hours involved are substantial: 7-9 hours per week on average for men and 3-4 hours for women.

A reduction in standard weekly hours increases overtime work, both incidence and hours; this is the case for women as well as for men. An increase in the wage rate decreases the incidence of overtime, but slightly increases overtime hours for those working overtime. These wage effects are stronger for men than for women. Union coverage emerges as of minor empirical importance in spite of its dramatic decline. The occupation and industry structure of employment have been changing away from high overtime jobs towards occupations and industries where less paid overtime is worked. Although many of these effects are statistically significant their practical effects tend to be relatively small. Taken together, these economic variables can explain almost half of the change in the incidence of overtime for men, and most of the change in overtime hours worked by women. However, they are less successful in explaining the change in overtime hours worked by men or the change in the incidence of overtime for women.

Our main analytical conclusion concerns the appropriate econometric model. We find evidence that the effects identified in previous studies may be seriously biased (Table 2 versus Table 3, and Table 4 versus Table 5). We conclude therefore that using an econometric framework controlling for both the censoring of overtime hours at zero and for the correlation between explanatory variables and unobserved job and individual characteristics is of major empirical importance.

REFERENCES

- Bauer, T. and Zimmermann, K.F. (1999) Overtime work and overtime compensation in Germany, Scottish Journal of Political Economy, 46, 419-436.
- Beatson, M. (1995) Labour Market Flexibility, Research Series no. 48, Department of Employment, London.
- Bell, D.N.F. and Hart, R.A. (1998) Working time in Great Britain, 1975-1994, Journal of the Royal Statistical Society, Series A, 161 (3), 327-348.
- Bell, D.N.F. and Hart, R.A. (1999) Unpaid work, Economica, 66, 271-290.
- Bell, D.N.F. and Hart, R.A. (1999a) Overtime working in an unregulated labor market, *IZA Discussion Paper*, no. 44, Institute for the Study of Labor, Bonn.
- Calmfors, L. and Hoel, M. (1988) Worksharing and overtime, Scandinavian Journal of Economics, 90 (1), 45-52.
- Contensou. F. and Vranceanu, R. (2000) Working time in a trade union model, chapter 10 in: Working *Time: Theory and Policy Implications*, Edward Elgar, Cheltenham.

Hamermesh, D.S. (1993) Labor Demand, Princeton University Press, Chichester, West-Sussex.

Hart, R.A. (1987) Working Time and Employment, Allen and Unwin, London.

Hausman, J.A. (1978) Specification tests in econometrics, Econometrica, 46 (6), 1251-1271.

- Heckman, J.J. (1976) The common structure of statistical models of truncation, sample selection and limited dependent variables and a simple estimator for such models, *Annals of Economic and Social Measurement*, 5, 475-492.
- Heckman, J.J. and Robb R. (1985) Using longitudinal data to estimate age, period and cohort effects in earnings equations, pp. 137-150 in W.M. Mason and S.E. Fienberg (eds.), Cohort Analysis in Social Research: Beyond the Identification Problem, Springer-Verlag: New York

Honoré, B.E. (2002) Nonlinear models with panel data, Portuguese Economic Journal, 1 (2), 163-179.

Hunt, J. (1999) Has work-sharing worked in Germany?, The Quarterly Journal of Economics, 117-148.

- Kalwij, A.S. (2003). A Maximum Likelihood Estimator based on first differences for a panel data Tobit model with individual specific effects, forthcoming, *Economics Letters*.
- Kalwij, A.S. (2003a). A two-step first-difference estimator for a panel data Tobit model, *working paper*, Department of Economics, Tilburg University.
- Kalwij, A.S. and Alessie, R. (2003) Permanent and transitory wage inequality of British men, 1975-2001: year, age and cohort effects, *working paper*, University of Amsterdam.
- Lilien, D.M. and Hall, R.E. (1986) Cyclical fluctuations in the labor market, Chapter 17 in Handbook of Labor Economics, Vol.II, editors: O.Ashenfelter and R. Layard, Elsevier Science Publishers, Amsterdam.
- Mundlak, Y. (1978) On the pooling of time series and cross section data, Econometrica, 46, 69-86.
- Nickell, S.J. (1983) Working hours in Britain, Working Paper no. 565, Centre for Labour Economics, London School of Economics.
- Pencavel, J. (1986) Labor supply of men: a survey, Chapter I in Handbook of Labor Economics, Vol.I, editors: O.Ashenfelter and R. Layard, Elsevier Science Publishers, Amsterdam.
- Rochina-Barrachina, M.E. (1999) A new estimator for panel data sample selection models, Annales d'Économie et de Statistique, 55/56, 153-181.
- Sallis, H. (1970) Overtime in electricity supply: its incidence and control in England and Wales 1954-1969, Occasional Paper, British Journal of Industrial Relations.
- Tobin, J. (1958) Estimation of relationships for limited dependent variables, Econometrica, 26, 24-36.
- Trejo, S.J. (1991) The effects of overtime pay regulation on worker compensation, American Economic Review, 81 (4), 719-740.
- Trejo, S.J. (1993) Overtime pay, overtime hours, and labor unions, Journal of Labor Economics, 11 (2), 253-278.
- Trejo, S.J. (2003) Does the statutory overtime premium discourage long workweeks?, Industrial and Labor Relations Review, 56 (3), 530-551.
- Wooldridge, J.M. (1995) Selection corrections for panel data models under conditional mean independence assumptions, *Journal of Econometrics*, 68, 115-132.

APPENDIX A:

	0500			. 1	Weekly paid working					
	OECD	employment	NESPD	raw data	NESPD fir	al sample	Female emplo	syment share	hou	Jrs
	Millian	(0/					NESPD		NESPD
	I*IIIIIOn Mar	s of persons	% (Mar	of employed	population	\ A /	OECD		OECD/LFS	tinai sampie
1075	Inten	vvomen	I*ien	vvomen	Men	vvomen	% 20.7	% 24.0	Hours	Hours
19/5	15.2	9.6	0.66	0.60	0.27	0.23	38.7	34.8	36.1	39.2
1976	15.0	9.5	0.71	0.67	0.37	0.33	38.9	36.6	35.8	38.6
1977	14.9	9.6	0.71	0.68	0.38	0.35	39.3	37.3	35.4	38.7
1978	14.9	9.8	0.71	0.68	0.37	0.34	39.5	37.6	35.1	38.7
1979	15.0	10.0	0.70	0.67	0.37	0.34	40.0	37.6	34.9	38.6
1980	14.9	10.1	0.70	0.68	0.38	0.35	40.2	38.1	34.0	38.1
1981	14.3	9.8	0.73	0.70	0.40	0.37	40.6	38.7	32.9	37.4
1982	13.9	9.7	0.72	0.70	0.41	0.38	41.0	39.2	33.2	37.3
1983	13.7	9.7	0.71	0.69	0.41	0.38	41.4	39.6	32.9	37.2
1984	13.9	9.9	0.69	0.67	0.40	0.37	41.6	39.9	33.3	37.4
1985	14.1	10.2	0.65	0.65	0.38	0.36	41.9	40.6	33.9	37.5
1986	14.0	10.4	0.67	0.65	0.39	0.36	42.5	40.5	33.9	37.5
1987	14.1	10.6	0.67	0.67	0.39	0.35	42.9	40.5	33.7	37.5
1988	14.7	11.0	0.66	0.68	0.39	0.35	42.9	40.7	34.5	37.8
1989	15.1	11.5	0.63	0.66	0.38	0.35	43.2	41.3	34.3	37.9
1990	15.2	11.6	0.64	0.67	0.37	0.36	43.3	42.3	34.0	37.7
1991	14.8	11.5	0.64	0.69	0.38	0.38	43.7	43.4	34.0	37.1
1992	14.2	11.5	0.63	0.68	0.38	0.38	44.7	44.3	33.2	36.9
1993	13.9	11.5	0.61	0.66	0.38	0.38	45.2	44.9	33.1	36.8
1994	14.1	11.5	0.61	0.67	0.39	0.38	45.I	44.9	33.4	36.9
1995	14.3	11.6	0.59	0.66	0.40	0.41	44.9	45.3	33.4	36.8
1996	14.4	11.8	0.58	0.65	0.40	0.41	45.I	45.5	33.4	36.8
1997	14.7	12.0	0.54	0.61	0.38	0.39	45.0	45.8	33.4	36.7
1998	14.9	12.1	0.56	0.62	0.37	0.39	44.9	45.6	33.3	36.6
1999	15.0	12.3	0.55	0.62	0.37	0.38	45.0	45.5	33.1	36.4
2000	15.2	12.4	0.52	0.59	0.36	0.38	45.0	46.3	32.8	36.1
2001	15.4	12.6	0.52	0.60	0.30	0.31	45.1	46.1	32.9	36.2

Table A2. A comparison of the raw and final samples from the NESPD with OECD statistics.

S		All birth	h-cohorts Birth-cohorts 1942-1954							
	I	Raw data	Selected	d sample	I	Raw data	Selecte	d sample		
	Men	Women	Men	Women	Men	Women	Men	Women		
	%	%	%	%	%	%	%	%		
0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0		
I	76.5	74.1	79.5	77.6	80. I	77.4	81.1	79.3		
2	61.6	58. I	58.9	55.3	66.9	63.0	62.2	58.6		
3	50.4	46.3	44.7	40.4	56.7	51.9	48.8	44.5		
4	41.7	37.3	34.5	29.9	48.4	43.2	38.9	34.2		
5	34.7	30.2	26.8	22.4	41.6	36.2	31.3	26.6		
6	28.9	24.6	21.0	16.8	35.8	30.4	25.4	20.7		
7	24.2	20.0	16.5	12.7	30.9	25.5	20.6	16.1		
8	20.2	16.3	13.0	9.6	26.6	21.4	16.9	12.6		
9	16.9	13.2	10.3	7.3	23.0	17.8	13.8	9.9		
10	14.1	10.7	8.I	5.5	19.7	14.8	11.3	7.7		
П	11.6	8.6	6.4	4.1	16.9	12.3	9.2	6.0		
12	9.6	6.8	5.0	3.1	14.3	10.1	7.5	4.6		
13	7.8	5.4	3.9	2.3	12.1	8.2	6.0	3.5		
14	6.4	4.3	3.0	1.7	10.2	6.7	4.8	2.7		
15	5.I	3.3	2.3	1.3	8.5	5.3	3.9	2.1		
16	4.1	2.6	1.7	0.9	7.0	4.2	3.0	١.5		
17	3.2	1.9	1.3	0.6	5.7	3.3	2.4	1.1		
18	2.5	1.5	1.0	0.4	4.6	2.6	1.8	0.8		
19	1.9	1.1	0.7	0.3	3.6	1.9	1.4	0.6		
20	1.4	0.8	0.5	0.2	2.8	1.4	1.0	0.4		
21	1.0	0.5	0.3	0.1	2.1	1.0	0.7	0.3		
22	0.7	0.4	0.2	0.1	1.5	0.7	0.5	0.2		
23	0.5	0.2	0.1	0.1	1.1	0.5	0.3	0.1		
24	0.3	0.1	0.1	0.0	0.7	0.3	0.2	0.1		
25	0.2	0.1	0.0	0.0	0.4	0.2	0.1	0.0		
26	0.1	0.0	0.0	0.0	0.2	0.1	0.1	0.0		

Table A3. Survivor rates in the sample, between periods t and t-s.

APPENDIX B: OVERTIME INCIDENCE, OVERTIME HOURS (IF WORKING OVERTIME), AND THE DISTRIBUTION OF EMPLOYMENT BY OCCUPATION, INDUSTRY AND REGION.

Table B1. By occupation

	Overtime	Overtime						
	incidence	hours						
	(%)	per week		Share	of emplo	oyment (%)	
Year			1975	1980	1985	1990	1995	2000
Panel A: Men								
Managers and administrators	11.8	6.0	7.1	8.8	9.0	10.8	16.0	18.3
Professional	16.7	6.0	11.8	12.3	13.4	13.6	12.1	11.9
Associate professional/ technical	23.6	5.9	7.4	8.4	9.3	9.9	9.3	9.5
Clerical and secretarial	38.8	6.4	10.5	9.5	10.0	9.7	10.2	10.6
Craft and related occupations	53.0	8.5	25.8	24.6	22.6	21.1	17.7	16.4
Personal and protective services	51.8	7.1	4.3	4.7	5.5	5.6	6. I	6.4
Sales	17.5	6.6	3.6	3.6	4.I	4.2	4.8	4.7
Plant and machine operatives	59.1	9.7	20.8	20.2	18.2	17.7	16.4	15.1
Other	56.5	9.7	8.7	8.0	7.9	7.5	7.3	7.2
All	39.4	8.2	100.0	100.0	100.0	100.0	100.0	100.0
Panel B: Women								
Managers and administrators	11.7	3.4	1.6	2.1	2.4	3.7	7.9	10.0
Professional	6.8	3.3	7.6	7.4	8.0	9.2	9.9	10.5
Associate professional/ technical	11.4	3.5	7.9	9.6	12.2	12.6	11.5	11.5
Clerical and secretarial	18.3	3.2	39.5	36.5	36.9	38. I	36.0	33.9
Craft and related occupations	16.7	4.5	4.9	4.2	3.5	3.1	2.4	١.7
Personal and protective services	15.6	4.5	7.6	8.9	9.1	8.3	10.0	11.8
Sales	23.4	4.3	8.0	8.4	8.6	9.2	9.1	10.4
Plant and machine operatives	23.6	4.9	9.7	8.5	6.4	5.9	4.8	3.8
Other	15.2	4.5	13.1	14.5	12.9	10.0	8.4	6.5
All	16.3	3.8	100.0	100.0	100.0	100.0	100.0	100.0

Table B2: By industry.

	Overtime	Overtime							
	incidence	hours	Shaws of any large of (%)			(0/)			
Year	(%)	per week	1975	Snare		loomen	t (%)	2000	
Panel A: Men			17/5	1760	1705	1770	1775	2000	
Agriculture	51.6	10.2	14	13	13	11	12	10	
Energy and water supply	47.4	76	54	5.2	53	4.6	3.0	2.0	
Mineral extractions, metal and chemical	74,7	7.0	5.4	5.2	5.5	4.0	5.0	2.0	
products	42.1	8.9	8.5	7.8	6.7	6.5	5.6	3.9	
Metal goods and engineering products	48.4	8.2	21.7	20.8	18.7	18.5	16.0	14.3	
Other manufacturing industries	43.7	9.0	12.2	11.5	10.8	10.8	11.1	9.8	
Construction	42.2	8.9	8.I	8.3	7.2	6.6	5.4	5.8	
Distributive trades and catering	31.8	7.3	8.2	10.0	10.9	11.5	13.5	15.2	
Transport and communications	56.0	10.0	12.5	11.3	10.8	10.6	10.1	9.9	
Financial services	24.6	5.6	4.9	6.0	7.7	10.6	12.8	15.6	
Public administration	33.0	6.2	9.8	9.2	9.7	9.2	8.9	8.3	
Educational services	12.0	6.4	2.7	3.5	3.8	3.0	4.6	6.4	
Health services	34.1	7.6	2.0	2.3	2.9	2.5	2.8	2.5	
Business and other services	29.1	7.5	2.4	2.9	4.2	4.6	5.1	5.2	
All	39.4	8.2	100.0	100.0	100.0	100.0	100.0	100.0	
Panel B: Women									
Agriculture	18.2	4.8	0.3	0.3	0.3	0.2	0.3	0.4	
Energy and water supply	20.0	3.6	1.8	1.6	1.6	١.5	1.0	0.7	
Mineral extractions, metal and chemical									
products	15.6	4.1	3.5	3.0	2.3	2.2	1.9	١.5	
Metal goods and engineering products	20.3	4.1	9.5	8.1	6.3	5.7	4.6	3.6	
Other manufacturing industries	16.7	4.4	12.7	10.6	8.7	8.2	7.1	5.2	
Construction	8.7	3.9	1.1	1.1	1.1	1.1	0.9	1.0	
Distributive trades and catering	20.2	4.2	15.6	17.3	16.9	17.4	17.4	18.4	
Transport and communications	23.5	4.3	4.6	3.8	3.6	3.9	3.5	3.3	
Financial services	25.9	2.6	8. I	9.0	11.2	14.5	15.3	17.5	
Public administration	12.8	3.8	10.5	9.7	9.5	9.9	10.3	8.7	
Educational services	6.9	3.9	13.3	13.8	12.7	10.1	12.2	16.6	
Health services	11.4	3.7	12.6	14.5	17.0	15.3	14.5	13.3	
Business and other services	15.1	4.7	6.4	7.2	8.8	9.9	11.0	9.9	
All	16.3	3.8	100.0	100.0	100.0	100.0	100.0	100.0	

Table B3. By region.

	Overtime	Overtime hours						
	incidence (%)	per week		Share	of emp	loyment	(%)	
			1975	1980	1985	1990	1995	2000
Panel A: Men								
Greater London	34.0	8.1	15.6	15.7	15.7	15.5	14.7	14.0
South-East (excluding Greater London)	39.1	8.2	14.8	15.9	16.3	16.6	17.0	18.1
East Anglia	42.8	8.3	3.4	3.3	3.4	3.6	4.0	4.5
South West	39.1	7.7	7.3	7.4	7.9	8. I	8.0	8.6
West Midlands	41.1	8.2	9.7	9.9	9.6	10.4	9.9	9.7
East Midlands	43.0	8.4	6.9	7.0	7.2	7.3	7.8	7.7
Yorkshire & Humberside	41.6	8.5	10.0	9.7	9.3	9.3	9.4	9.4
North West	39.9	8.2	12.1	12.0	11.8	11.3	11.2	10.6
North	40.5	8.4	4.4	4.7	4.5	4.I	4.1	4.I
Wales	37.9	8.2	5.0	4.7	4.6	4.6	4.6	4.4
Scotland	40. I	8.4	10.8	9.9	9.8	9.2	9.4	8.7
All	39.4	8.2	100.0	100.0	100.0	100.0	100.0	100.0
Panel B: Women								
Greater London	15.1	3.6	16.9	15.8	15.3	14.9	14.2	13.7
South-East (excluding Greater London)	17.6	3.7	15.0	16.3	16.7	17.4	17.4	18.1
East Anglia	17.8	3.7	2.8	2.9	3.1	3.2	3.6	4.3
South West	17.7	3.6	6.8	7.1	7.7	8.3	8.5	8.9
West Midlands	15.9	3.9	9.6	10.2	9.6	9.8	9.6	9.4
East Midlands	17.5	3.8	6.2	6.I	6.6	6.6	7.2	7.2
Yorkshire & Humberside	16.7	3.9	9.5	9.4	9.1	9.0	9.1	9.3
North West	16.1	4.0	12.4	12.5	12.4	11.8	11.6	10.9
North	15.5	3.9	4.4	4.5	4.3	4.2	4.2	4.1
Wales	14.5	4.1	4.4	4.I	4.5	4.4	4.3	4.7
Scotland	15.2	3.9	11.9	11.2	10.7	10.3	10.1	9.3
ΑΙΙ	16.3	3.8	100.0	100.0	100.0	100.0	100.0	100.0

Recent publications of the Amsterdam Institute for Advanced Labour Studies

WORKING PAPERS

- 03-25 "Wage Indicator" Dataset Loonwijzer Januari 2004 dr Kea Tijdens
- 03-24 "Codeboek DUCADAM Dataset" December 2003 Drs Kilian Schreuder & dr Kea Tijdens
- 03-23 "Household Consumption and Savings Around the Time of Births and the Role of Education"December 2003 Adriaan S. Kalwij
- 03-22 "A panel data analysis of the effects of wages, standard hours and unionisation on paid overtime work in Britain" October 2003 Adriaan S. Kalwij
- 03-21 "A Two-Step First-Difference Estimator for a Panel Data Tobit Model" December 2003 Adriaan S. Kalwij
- 03-20 "Individuals' Unemployment Durations over the Business Cycle" June 2003 Adriaan S. Kalwij
- 03-19 "Een onderzoek naar CAO-afspraken op basis van de FNV cao-databank en de AWVN-database" December 2003 Kea Tijdens & Maarten van Klaveren
- 03-18 "Permanent and Transitory Wage Inequality of British Men, 1975-2001: Year, Age and Cohort Effects"
 October 2003 Adriaan S. Kalwij & Rob Alessie
- 03-17 "Working Women's Choices for Domestic Help" October 2003 Kea Tijdens, Tanja van der Lippe & Esther de Ruijter
- 03-16 "De invloed van de Wet arbeid en zorg op verlofregelingen in CAO's" October 2003 Marieke van Essen
- 03-15 "Flexibility and Social Protection" August 2003 Dr Ton Wilthagen
- 03-14 "Top Incomes in the Netherlands and The United Kingdom over the Twentieth Century" September 2003 Sir dr. A.B.Atkinson and dr. W. Salverda
- 03-13 "Tax Evasion in Albania: an Institutional Vacuum" April 2003 dr. Klarita Gërxhani
- 03-12 "Politico-Economic Institutions and the Informal Sector in Albania" May 2003 dr. Klarita Gërxhani
- 03-11 "Tax Evasion and the Source of Income: An experimental study in Albania and the Netherlands" May 2003 dr. Klarita Gërxhani
- 03-10 "Chances and limitations of "benchmarking" in the reform of welfare state structures the case of pension policy"
 May 2003 dr. Martin Schludi
- 03-09 "Dealing with the "flexibility-security-nexus: Institutions, strategies, opportunities and barriers" May 2003 prof. Ton Wilthagen en dr. Frank Tros
- 03-08 "Tax Evasion in Transition: Outcome of an Institutional Clash -Testing Feige's Conjecture" March 2003 dr. Klarita Gërxhani
- 03-07 "Teleworking Policies of Organisations- The Dutch Experiencee" February 2003 dr. Kea Tijdens en Maarten van Klaveren
- 03-06 "Flexible Work- Arrangements and the Quality of Life" February 2003 drs. Cees Nierop

- 01-05 Employer's and employees' preferences for working time reduction and working time differentiation

 A study of the 36 hours working week in the Dutch banking industry"
 2001 dr. Kea Tijdens
- 01-04 "Pattern Persistence in Europan Trade Union Density" October 2001 prof. dr. Danielle Checchi, prof. dr. Jelle Visser
- 01-03 "Negotiated flexibility in working time and labour market transitions The case of the Netherlands" 2001 prof. dr. Jelle Visser
- 01-02 "Substitution or Segregation: Explaining the Gender Composition in Dutch Manufacturing Industry 1899 – 1998" June 2001 Maarten van Klaveren – STZ Advies en Onderzoek , Eindhoven, dr. Kea Tijdens
- 00-01 "The first part-time economy in the world. Does it work?" June 2000 prof. dr. Jelle Visser

RESEARCH REPORTS

- 02-17 "Industrial Relations in the Transport Sector in the Netherlands" December 2002 dr. Marc van der Meer & drs. Hester Benedictus
- 03-16 "Public Sector Industrial Relations in the Netherlands: framework, principles, players and Representativity" January 2003 drs. Chris Moll, dr. Marc van der Meer & prof.dr. Jelle Visser
- 02-15 "Employees' Preferences for more or fewer Working Hours: The Effects of Usual, Contractual and Standard Working Time, Family Phase and Household Characteristics and Job Satisfaction" December 2002 dr. Kea Tijdens
- 02-13 "Ethnic and Gender Wage Differentials An exploration of LOONWIJZERS 2001/2002" October 2002 dr. Aslan Zorlu
- 02-12 "Emancipatie-effectrapportage belastingen en premies een verkenning naar nieuwe mogelijkheden vanuit het belastingstelsel 2001" August 2002 dr. Kea Tijdens, dr. Hettie A. Pott-Buter
- 02-11 "Competenties van Werknemers in de Informatiemaatschappij Een survey over ICT-gebruik" June 2002 dr. Kea Tijdens & Bram Steijn
- 02-10 "Loonwijzers 2001/2002. Werk, Ionen en beroepen van mannen en vrouwen in Nederland" June 2002 Kea Tijdens, Anna Dragstra, Dirk Dragstra, Maarten van Klaveren, Paulien Osse, Cecile Wetzels, Aslan Zorlu
- 01-09 "Beloningsvergelijking tussen markt en publieke sector: methodische kantekeningen" November 2001 Wiemer Salverda, Cees Nierop en Peter Mühlau
- 01-08 "Werken in de Digitale Delta. Een vragenbank voor ICT-gebruik in organisaties" June 2001 dr. Kea Tijdens
- 01-07 "De vrouwenloonwijzer. Werk, lonen en beroepen van vrouwen." June 2001 dr. Kea Tijdens
- 00-06 "Wie kan en wie wil telewerken?" Een onderzoek naar de factoren die de mogelijkheid tot en de behoefte aan telewerken van werknemers beïnvloeden." November 2000 dr. Kea Tijdens, dr. Cecile Wetzels en Maarten van Klaveren
- 00-05 "Flexibele regels: Een onderzoek naar de relatie tussen CAO-afspraken en het bedrijfsbeleid over flexibilisering van de arbeid." Juni 2000 dr. Kea Tijdens & dr. Marc van der Meer
- 00-04 "Vraag en aanbod van huishoudelijke diensten in Nederland" June 2000 dr. Kea Tijdens
- 00-03 "Keuzemogelijkheden in CAO's" June 2000 Caroline van den Brekel en Kea Tijdens
- 00-02 "The toelating van vluchtelingen in Nederland en hun integratie op de arbeidsmarkt." Juni 2000 Marloes Mattheijer
- 00-01 "The trade-off between competitiveness and employment in collective bargaining: the national consultation process and four cases of enterprise bargaining in the Netherlands" Juni 2000 Marc van der Meer (ed), Adriaan van Liempt, Kea Tijdens, Martijn van Velzen, Jelle Visser

AIAS

AIAS is a young interdisciplinary institute, established in 1998, aiming to become the leading expert centre in the Netherlands for research on industrial relations, organisation of work, wage formation and labour market inequalities.

As a network organisation, AIAS brings together high-level expertise at the University of Amsterdam from five disciplines:

- Law
- Economics
- Sociology
- Psychology
- Health and safety studies

AIAS provides both teaching and research. On the teaching side it offers a Masters in Advanced Labour Studies/Human Resources and special courses in co-operation with other organizations such as the National Trade Union Museum and the Netherlands Institute of International Relations 'Clingendael'. The teaching is in Dutch but AIAS is currently developing a MPhil in Organisation and Management Studies and a European Scientific Master programme in Labour Studies in co-operation with sister institutes from other countries.

AIAS has an extensive research program (2000-2004) building on the research performed by its member scholars. Current research themes effectively include:

- The impact of the Euro on wage formation, social policy and industrial relations
- Transitional labour markets and the flexibility and security trade-off in social and labour market regulation
- The prospects and policies of 'overcoming marginalisation' in employment
- The cycles of policy learning and mimicking in labour market reforms in Europe
- Female agency and collective bargaining outcomes
- The projects of the **LoWER** network.



AMSTERDAMS INSTITUUT VOOR ARBEIDSSTUDIES Universiteit van Amsterdam

Plantage Muidergracht 4 1018 TV Amsterdam the Netherlands tel +31 20 525 4199 fax +31 20 525 4301 aias@uva.nl www.uva-aias.net